Author’s response to reviews

Title: The 3-hydroxy-3-methylglutaryl-coenzyme A reductase inhibitors, simvastatin, lovastatin and mevastatin inhibit proliferation and invasion of melanoma cells.

Authors:

Sharon A Glynn (glynnsh@mail.nih.gov)
Dermot O’Sullivan (dermot.osullivan2@mail.dcu.ie)
Alex J Eustace (alex.eustace@dcu.ie)
Martin Clynes (martin.clynes@dcu.ie)
Norma O’Donovan (norma.odonovan@dcu.ie)

Version: 5 Date: 10 December 2007

Author’s response to reviews: see over
Mona Jazayeri  
Assistant Editor  
BMC-series Journals  

10th Dec 2007  

Dear Dr. Jazayeri,  

Thank you for your email of 7th Dec 2007. We are delighted that you are now willing to accept our manuscript (MS-1614590434152778) entitled “The 3-hydroxy-3-methylglutaryl-coenzyme A reductase inhibitors, simvastatin, lovastatin and mevastatin inhibit proliferation and invasion of melanoma cells” for publication in the *BMC Cancer*.  

We have revised the manuscript according to the reviewer’s suggestion. The p values are incorporated in the figures as symbols and the supplemental tables have been removed. We think the revisions have improved the quality of the manuscript and believe that it is now ready for publication in *BMC Cancer*.  

Thank you for your time and consideration.  

Yours sincerely,  

Norma O’Donovan, Ph.D. and Sharon Glynn, Ph.D., M.PH.  
National Institute for Cellular Biotechnology  
Dublin City University  
Glasnevin, Dublin 9, Ireland.  
Phone +353-1-7005700  
Email: norma.odonovan@dcu.ie, glynnsh@mail.nih.gov